Author’s response to reviews

Title: What have we learned about communication inequalities during the H1N1 pandemic? A systematic review of the literature

Authors:

Leesa Lin (llin@hsph.harvard.edu)
Elena Savoia (esavoia@hsph.harvard.edu)
Foluso Agboola (fagboola@hsph.harvard.edu)
Kasisomayajula Viswanath (vish_viswanath@dfci.harvard.edu)

Version: 3 Date: 23 October 2013

Author’s response to reviews:

Boston, October 23, 2013
Re: Manuscript Submission

Dear Editor,

I am submitting, on behalf of all authors, the manuscript entitled “What have we learned about communication inequalities during the H1N1 pandemic? A systematic review of the literature” to be considered for publication as a Research Article of the journal, BMC Public Health (“Health Behavior, Health Promotion and Society” Section).

The article examines the extent to which communications in Public Health Emergency Preparedness (PHEP) has received attention by the scientific community during the response to the recent H1N1 pandemic. It describes current research in the area of communications in PHEP focusing on communication inequalities and their links to specific communications and preparedness outcomes. We believe our manuscript is important in this area and the findings could help effectively achieve communication and preparedness outcomes and provide useful information to the practitioners developing communication messages during the response to a public health emergency.

The material has not been published nor is being considered for publication elsewhere. The authors declare no conflict of interest in the conduction or production of any part of the work presented in this manuscript. The authors deny any similar or related work submitted or published elsewhere.

I look forward to hearing from your team.

Best regards,
Leesa Lin, MSPH
Corresponding author,
llin@hsph.harvard.edu
phone 617-384-9872